Dear Editor:

Thank you for your comments concerning our manuscript entitled “Association between osteoporosis and benign paroxysmal positional vertigo: a systematic review”. I think that is very important for the enhancement of our manuscript. The following are our responses to your comments.

Response to comments of reviewer 1:

1. We recommend that you copyedit the paper to improve the style of written English. If this is not possible, you may need to use a professional language editing service. For authors who wish to have the language in their manuscript edited by a native-English speaker with scientific expertise, BioMed Central recommends Edanz (www.edanzediting.com/bmc1). BioMed Central has negotiated a 10% discount to the fee charged to BioMed Central authors by Edanz. Use of an editing service is neither a requirement nor a guarantee of acceptance for publication. - Please include the email addresses of all authors in the title page

Response: the paper has been edited by Edanz.

2. Please use authors' initials in author's contribution statement

Response: That has been revised.

3. Please place tables after the references

Response: That has been revised.

Thank you and best regards.

Yours sincerely,

Qirong Wang